Author's response to reviews

Title: Health Insurance, Neighborhood Income, and Emergency Department Usage by Utah Children 1996-1998

Authors:

Anthony Suruda (tonysuruda@aol.com)
Thomas Burns (tburns@ou.edu)
Stacey Knight (Stacey.Knight@hsc.utah.edu)
Jonathan M Dean (Mike.Dean@hsc.utah.edu)

Version: 2 Date: 14 January 2005

Author's response to reviews: see over